Author's response to reviews

Title: GSTT2 promoter polymorphisms and colorectal cancer risk

Authors:

Sang-Geun Jang (jangsg@gmail.com)
Il-Jin Kim (iljinkim@hotmail.com)
Hio Chung Kang (virgo@ncc.re.kr)
Hye-Won Park (pearljuno@naver.com)
Sun-A Ahn (biosun7942@hanmail.net)
Hyun-Ju Yoon (ynhi31@nate.com)
Kun Kim (huntkool@hotmail.com)
Hai-Rim Shin (hrshin@ncc.re.kr)
Jin Soo Lee (jslee@ncc.re.kr)
Jae-Gahb Park (jgpark@plaza.snu.ac.kr)

Version: 3 Date: 22 January 2007

Author's response to reviews: see over
Jan 20, 2007

Dear Dr. Scott Edmunds

Manuscript number: 1711548218119934
Manuscript title: GSTT2 promoter polymorphisms and colorectal cancer risk

I would like to appreciate you that my manuscript was accepted for publication in BMC Cancer. Dr. Stefano Landi was recommended to change the sentences like "strong linkage disequilibrium" with "good level of LD" or something similar and the BMC editorial production team was suggested that website addresses must be removed from text. We changed "strong linkage disequilibrium" with "good level of LD" in line 3rd, page 11 and line 3rd, page 12 and removed website addresses (http://www.stat.washington.edu/stephens/software.html and http://www.ngic.re.kr:8080/khapmap/khap_product/hardy.jsp), in 12th and 16th lines, page 9.

Thank you for your consideration for this manuscript.

Sincerely yours,

Jae-Gahb Park, M.D., Ph.D.
Laboratory of Cell Biology
Cancer Research Institute and Cancer Research Center
Seoul National University, Seoul, 110-799, Korea
E-mail: jgpark@plaza.snu.ac.kr
Fax: +82-2-742-4727